Author’s response to reviews

Title: Sexual behavior and drug consumption among young adults in a shantytown in Lima, Peru

Authors:

Juan A Gálvez-Buccollini (galvezbuccollini@yahoo.com)
Suzanne DeLea (sdelea@salud.unm.edu)
Phabiola M Herrera (pherrera@prisma.org.pe)
Robert H Gilman (rgilman@jhsph.edu)
Valerie Paz-Soldan (vpazsold@tulane.edu)

Version: 7  Date: 16 January 2009

Author’s response to reviews:

Dear Editor,

1. First page in corresponding author said: "autor", changed by "author"
2. In the first sentence of the abstract said: "Risky sexual behaviors of young adults have received increasing attention during the last decade", changed by "Risky sexual behaviors of young adults have received increasing attention during the last decades"
3. In methods, section design, last sentence first paragraph had a colon and semicolon together after the word "alcohol and drug use,;". The colon have been deleted.
4. The title of the table 2 in the draft, that you sent me Jan 7, is not in the right place. It is under the table and It should be above the table.

Thanks for your assistance with these changes,

Juan A Galvez-Buccollini M.D.